Author’s response to reviews

Title: Anxiety, depression and relationship satisfaction in the pregnancy following stillbirth and after the birth of a live-born baby: A prospective study

Authors:

Ida Kathrine Gravensteen (ida.gravensteen@gmail.com)

Eva-Marie Jacobsen (e.m.jacobsen@medisin.uio.no)

Per Morten Sandset (p.m.sandset@medisin.uio.no)

Linda Helgadottir (uxliot@ous-hf.no)

Ingela Rådestad (ingela.radestad@shh.se)

Leiv Sandvik (UXLEDV@ous-hf.no)

Øivind Ekeberg (UXOIEK@ous-hf.no)

Version: 2 Date: 07 Jan 2018

Author’s response to reviews:

Dear editor,

We hereby submit our revised manuscript “Anxiety, depression and relationship satisfaction in the pregnancy following stillbirth and after the birth of a live-born baby: A prospective study” for consideration of publication in BMC Pregnancy and Childbirth. When preparing our revised manuscript, we have done our best to comply with the remarks provided by the editor and the reviewers. Each comment has been given a response in the following point-to-point listing, and the manuscript and additional files has been revised accordingly. All listed authors have approved the final version of this manuscript for submission.

On behalf of all the authors,

Ida Kathrine Gravensteen

MD, PhD
Technical Comments:

1) Figure files should contain only the image/graphic, as well as any associated keys/annotations. If titles/legends are present within the figure files, please remove them.

Response: The titles has been removed from the figure files and can be found at the end of the manuscript (Figure legends, pages 33-34, lines 685-710)

2) At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: A clean version of the final manuscript is hereby uploaded.